Author's response to reviews

Title: Low levels of IgM antibodies to oxidized cardiolipin increase and high levels decrease risk of cardiovascular disease among 60-year olds: a prospective study.

Authors:

Jun Su (jun.su@ki.se)
Xiang Hua (xiang.hua@ki.se)
Max Vikstrom (max.vikstrom@ki.se)
Karin Leander (karin.leander@ki.se)
Bruna Gigante (bruna.gigante@ki.se)
Mai-Lis Hellenius (mai-lis.hellenus@ki.se)
Ulf de Faire (ulf.defaire@ki.se)
Johan Frostegard (johan.frostegard@ki.se)

Version: 5 Date: 2 November 2012

Author's response to reviews:

We have now revised the manuscript as suggested, with keywords after abstract, and addition of reason to invite 60-year olds, in Methods section, 1st paragraph.